CSCI 1323 (Discrete Structures), Spring 2002 Homework 8

Assigned: April 25, 2002. Due: May 8, 2002. Credit: Not for credit.

1 Problems

Doing the following problems should help you review material from chapter 5. You are not required to turn any of them in. Answers to most can be found in the back of the textbook.

1. Do problem 2 on p. 351 of the textbook. 2. Do problem 31 on p. 356 of the textbook. 3. Do problem 39 on p. 357 of the textbook. 4. Do problem 48 on p. 358 of the textbook. 5. Do problem 66 on p. 360 of the textbook. 6. Do problem 1 on p. 373 of the textbook. 7. Do problem 16 on p. 377 of the textbook. 8. Do problem 36 on p. 378 of the textbook.